Author's response to reviews

Title: Renal effects of dexmedetomidine during coronary artery bypass surgery: a randomized placebo-controlled study

Authors:

    Kari Leino (kari.leino@tyks.fi)
    Markku Hynynen (markku.hynynen@hus.fi)
    Jouko Jalonen (jouko.jalonen@utu.fi)
    Markku Salmenperä (markku.salmenpera@hus.fi)
    Harry Scheinin (harry.scheinin@utu.fi)
    Riku Aantaa (riku.aantaa@tyks.fi)
    for the Dexmedetomidine in Cardiac Surgery Study Group
    (anne.kuitunen@kolumbus.fi)

Version: 6 Date: 8 May 2011

Author's response to reviews:

Dear Editorial TEAM,

We have now copied the abstract from the manuscript to the submission system. Now they match.

Figure 3 has been provided with a short overall title.

Figure 4 is now in .tif -format.

In addition, we found two mistakes in figure 2: In Y-axis seconds are replaced with minutes, as this is more appropriate and used also elsewhere in the results section. In addition, in X-axis the third time period is now corrected (24-48 h, not 24-28 h). We have replaced seconds with minutes also in Y-axis of figure 3 panel A.

Sincerely yours,

Kari Leino